Author's response to reviews

Title: Efficacy and Safety of Intra-articular Injections of Sodium Hyaluronate (Hyalgan) in the Treatment of Osteoarthritis of the Knee A Randomized, Controlled Trial in the Asian Population

Authors:

Teng-Le Huang (h.tengle@gmail.com)
Chi-Ching Chang (cchang@tmu.edu.tw)
Chian-Her Lee (chianherlee@yahoo.com.tw)
Shih-Ching Chen (csc@tmu.edu.tw)
Chien-Hung Lai (lch@tmu.edu.tw)
Ching-Lin Tsai (d95548008@ntu.edu.tw)

Version: 3 Date: 1 April 2011

Author's response to reviews: see over
Dear Editor-in-Chief and Editorial Assistant Judith Gorton,

Thank you very much for the e-mail of Mar. 21\textsuperscript{th}. We are very glad to have the guidance and information in revising this manuscript. Your comments are both helpful and constructive to us. Based on \url{http://www.icmje.org/faq_clinical.html}, we made registration on ClinicalTrials.gov, the attachment in the “Additional material files” is the related information. We also listed the trial registry and the unique identifying number, i.e.: NCT01319461, in the last section of the Abstract.

Thanks once more for your time and kindness. We look forward to the further information after your reviews.

Very Sincerely,

Ching-Lin Tsai, MD. PhD.
Professor
Department of Orthopedics, National Taiwan University Hospital, 7 Chung Shan South Road, Taipei, Taiwan, R.O.C. Tel:886 (02) 2312 3456 Ext. 5272
E-mail: d95548008@ntu.edu.tw